

Brussels, 29.11.2023 COM(2023) 752 final 2023/0436 (COD)

## Proposal for a

# REGULATION OF THE EUROPEAN PARLIAMENT AND OF THE COUNCIL

on passenger rights in the context of multimodal journeys

(Text with EEA relevance)

 $\{ SEC(2023) \ 392 \ final \} - \{ SWD(2023) \ 386 \ final \} - \{ SWD(2023) \ 387 \ final \} - \{ SWD(2023) \ 389 \ final \}$ 

EN EN

## EXPLANATORY MEMORANDUM

#### 1. CONTEXT OF THE PROPOSAL

## Reasons for and objectives of the proposal

The Sustainable and Smart Mobility Strategy confirms the ambition of the European Green Deal to achieve a 90% reduction in transport emissions by 2050. A key element of making transport more sustainable is achieving effective multimodality, where passengers who wish to travel long distances can do so by combining several modes of collective transport, thereby using the most sustainable and efficient mode for each leg of the journey. Protecting passengers along the way, in particular during travel disruptions, is key to enhance the attractiveness of such multimodal travels. Moreover, a failure to provide this protection could itself affect market growth for multimodal travel, with some passengers choosing to travel by car instead of collective transport (judging car travel to be both more convenient and more reliable).

The Sustainable and Smart Mobility Strategy announced measures to strengthen the regulatory framework for passenger rights. In that context, it observed that EU passenger rights should be better implemented and clearer for both carriers and passengers and that they should offer adequate assistance, reimbursement, possibly compensation when disruptions arise, and appropriate sanctions if the rules are not properly applied. The Commission announced that it will consider options and benefits to go further with a multimodal framework for passenger rights that is simplified, more consistent and harmonised.

Multimodal journeys occur when passengers combine at least two collective transport modes to reach a final destination, such as a flight with a rail service, or a rail service with a coach service. Firstly, multimodal travel typically involves travelling with a number of separate tickets that passengers buy individually ('category C' tickets). In addition, some intermediaries bundle such separate tickets into a multimodal product on their own initiative and sell it as such to passengers in one single commercial transaction ('category B' tickets). Finally, a limited number of carriers also offer multimodal journeys under a single contract of carriage ('category A' ticket). It is estimated that 91 million passengers performed multimodal journeys in 2019, and the number is expected to grow to 103.6 million in 2030 and 150.9 million in 2050. Expressed as share of the total number of passengers, multimodal passengers are estimated to increase from 0.7% in 2019 to 0.8% by 2050.

Today, while passengers who travel with only one collective transport mode (i.e. only by air, rail, bus or ship) enjoy rights in the event of travel disruptions,<sup>2</sup> they are not entitled to the

\_

Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions - Sustainable and Smart Mobility Strategy – putting European transport on track for the future, COM(2020) 789 final, 9.12.2020.

Regulation (EC) No 261/2004 of the European Parliament and of the Council of 11 February 2004 establishing common rules on compensation and assistance to passengers in the event of denied boarding and of cancellation or long delay of flights, and repealing Regulation (EEC) No 295/91 (OJ L 46, 17.2.2004, p. 1); Regulation (EC) No 1107/2006 of the European Parliament and of the Council of 5 July 2006 concerning the rights of disabled persons and persons with reduced mobility when travelling by (OJ L 204, 26.7.2006, p. 1); Regulation (EU) 2021/782 of the European Parliament and of the Council of 29 April 2021 on rail passengers' rights and obligations (recast) (OJ L 172, 17.5.2021, p. 1); Regulation (EU) No 1177/2010 of the European Parliament and of the Council of 24 November 2010 concerning the rights of passengers when travelling by sea and inland waterway and amending Regulation (EC) No 2006/2004 (OJ L 334, 17.12.2010, p. 1); Regulation (EU) No 181/2011 of the

same rights when switching to another mode as part of their journey. This implies that there is also no clear framework for determining the respective obligations and liabilities of the different travel service providers involved in a multimodal journey. National legal frameworks generally do not include provisions covering multimodal transport either. In addition, the current offer of travel insurances for multimodal journeys remains limited. The rights of passengers in the context of multimodal journeys depend therefore on the terms and conditions of the specific contract(s) of carriage.

Consequently, passengers lack information on the extent of their rights before and during multimodal travel and are not given information in real-time on possible travel disruptions and security alerts when they need to switch between modes. Moreover, they can possibly suffer a different treatment with regard to contract conditions and tariffs for multimodal journeys on the basis of their nationality or of the place of establishment of the carrier or intermediary. In addition, those passengers do not receive assistance (e.g. reimbursement, rerouting, accommodation, meals and refreshments) during their journey in the event of a travel disruption that occurs when switching between transport modes. Furthermore, passengers have difficulties to complain to carriers and other possible relevant actors (e.g. terminal operators, ticket vendors) about the lack of information or assistance, nor do they have clarity on which national authority to contact in such cases. This lack of a specifically designated authority also leads to uncertainty on the enforcement of information to and assistance of passengers during multimodal travel. The consequences of the lack of information and assistance to passengers when switching modes include loss of connections and money for citizens and biased decisions from citizens towards less sustainable/climate and environmentally friendly solutions.

The lack of rules for multimodal journeys also means that persons with disabilities and persons with reduced mobility (PRM) are not entitled to any particular assistance under EU law when transferring between transport modes, including at multimodal connecting points such as air-rail hubs. In the absence of such assistance, they will not be able to benefit from a seamless travel experience similar to that of other passengers.

Therefore, the objectives of this proposal are to ensure:

- non-discrimination between passengers with regard to transport conditions and the provision of tickets;
- minimum and accurate information to be provided to passengers in an accessible format and in a timely manner to passengers;
- passengers' rights in the event of disruption, in particular in the context of a missed connection between different transport modes;
- non-discrimination against, and assistance for, persons with disabilities and persons with reduced mobility;
- the definition and monitoring of service quality standards;
- the handling of complaints;
- general rules on enforcement.

European Parliament and of the Council of 16 February 2011 concerning the rights of passengers in bus and coach transport and amending Regulation (EC) No 2006/2004 (OJ L 55 28.2.2011, p. 1).

This initiative is part of the Commission Work Programme 2023 under Annex II (REFIT initiatives), under the heading 'An Economy that Works for People'.

## Consistency with existing policy provisions in the policy area

The Union law on passenger rights already raises the standards of protection for passengers when they travel by air, rail, sea and inland waterways, and bus and coach. In particular, these Regulations contain provisions on information and assistance to passengers which apply to each of these transport modes considered separately. This proposal aims to complement these existing rules by ensuring that passengers enjoy a similar level of protection where they switch between these transport modes during a journey.

This proposal is consistent with the proposal amending Regulations (EC) No 261/2004, (EC) No 1107/2006, (EU) No 1177/20, (EU) No 181/2011 and (EU)2021/782 as regards enforcement of passenger rights in the Union.<sup>3</sup> The rules on better application and enforcement of Union rules on passenger rights in this Regulation are fully aligned with that Regulation.

## • Consistency with other Union policies

This proposal is consistent with the Commission Delegated Regulation (EU) 2017/1926 on Multimodal Travel Information Services (MMTIS).<sup>4</sup> This Regulation requires data holders (e.g. transport authorities, transport operators and infrastructure managers) to make data on transport networks (including data on transport services such as timetables and hours of operation) accessible to data users, via national access points established by the Member States, when the data is available in a digital machine-readable format. It is also consistent with the proposal for the revision of the Commission Delegated Regulation (EU) 2017/1926<sup>5</sup>. This revision extends this requirement by gradually adding, from December 2025 to December 2028, dynamic data types such as real-time arrival and departure times. This is of essential importance for this proposal regarding real-time information to passengers about travel disruptions when switching modes. This will be further supported by the proposed amendment of the ITS Directive.<sup>6</sup> This amendment should ensure that ITS applications in the field of road transport enable seamless integration with other modes of transport, such as rail or active mobility, thus facilitating a shift to those modes whenever possible, to improve efficiency and accessibility.

This proposal is also consistent with the Communication on a common European mobility data space (EMDS), which outlines the proposed way forward for the creation of a common European mobility data space, including its objectives, main features, supporting measures and milestones. Its objective is to facilitate access, pooling and sharing of data from existing

\_

<sup>&</sup>lt;sup>3</sup> COM(2023)753 final of 29.11.2023.

Commission Delegated Regulation (EU) 2017/1926 of 31 May 2017 supplementing Directive 2010/40/EU of the European Parliament and of the Council with regard to the provision of EU-wide multimodal travel information services (OJ L 272, 21.10.2017, p. 1).

https://ec.europa.eu/info/law/better-regulation/have-your-say/initiatives/12912-EU-wide-multimodal-travel-new-specifications-for-information-services\_en.

Proposal for a Directive amending Directive 2010/40/EU on the framework for the deployment of Intelligent Transport Systems in the field of road transport and for interfaces with other modes of transport, COM(2021)813 final - 2021/0419 (COD) - (EP and Council agreed on the provisional text in June 2023).

and future transport and mobility data sources.<sup>7</sup> With regard to the exchange of real-time information in a business-to-business context, this proposal is also relevant in the context of the proposal for a Data Act, which addresses important aspects of data sharing, such as compensation, dispute settlement or technical protection measures.<sup>8</sup> The proposal on establishing a framework for a European Digital Identity could also play a role where it introduces European Digital Identity Wallets.<sup>9</sup> In light of the benefits in terms of security, convenience and accessibility, Member States should encourage the use of European Digital Identity Wallets for identification and authentication in multimodal transport scenarios, particularly aiding vulnerable persons or persons with disabilities.

Moreover, it should be clarified that the rules on package travel under Directive (EU)  $2015/2302^{10}$  and the proposal to revise it, <sup>11</sup> are also consistent with this proposal. Albeit both deal with travel, they concern two separate markets. Whereas this proposal – in line with the existing legislation on passenger rights and the proposal to amend these <sup>12</sup>— exclusively deals with passenger transport services (e.g. a combination of a rail and bus service), the rules on package travel cover a combination of different travel services offered by an organiser (e.g. a package combining a flight and hotel accommodation). This proposal also clarifies that where a right to reimbursement arises under Directive (EU) 2015/2302, it should apply instead of this Regulation with regard to single multimodal contracts. Moreover, it specifies that this Directive should apply where an organiser combines transport services for the purpose of a multimodal journey as part of a package.

## 2. LEGAL BASIS, SUBSIDIARITY AND PROPORTIONALITY

## Legal basis

Article 91(1) of the Treaty on the Functioning of the European Union (TFEU) serves as the legal basis for the adoption of EU legislation related to a common transport policy. Furthermore, pursuant to Article 100(2) TFEU the Union legislator may lay down appropriate provisions for sea and air transport. The proposal is based on Article 91(1) and Article 100(2) TFEU.

\_

Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions – Creation of a common European mobility data space, COM(2023)751 final of 29.11.2023.

Proposal for a Regulation of the European Parliament and of the Council on harmonised rules on fair access to and use of data (Data Act), COM(2023)68 final of 23.2.2022.

Proposal for a Regulation of the European Parliament and of the Council amending Regulation (EU) No 910/2014 as regards establishing a framework for a European Digital Identity, COM(2021)281 of 3.6.2021.

Directive (EU) 2015/2302 of the European Parliament and of the Council of 25 November 2015 on package travel and linked travel arrangements, amending Regulation (EC) No 2006/2004 and Directive 2011/83/EU of the European Parliament and of the Council and repealing Council Directive 90/314/EEC (OJ L 326, 11.12.2015, p. 1).

Proposal for a Directive of the European Parliament and of the Council amending Directive (EU) 2015/2302 to make the protection of travellers more effective and to simplify and clarify certain aspects of the Directive, COM(2023)905 final of 29.11.2023.

Proposal amending Regulations (EC) No 261/2004, (EC) No 1107/2006, (EU) No 1177/20, (EU) No 181/2011 and (EU)2021/782 as regards enforcement of passenger rights in the Union, COM(2023)753 final of 29.11.2023.

## • Subsidiarity (for non-exclusive competence)

Passenger rights for travel by air, rail, bus and coach (for long distance journeys), sea and inland waterways are already enshrined in EU law, <sup>13</sup> and it is only urban/short-distance passenger transport that has been mostly left within the jurisdictions of Member States. The current proposal seeks to address the lack of rights for passengers who travel using a combination of these modes. Without harmonisation of the rules protecting passengers during such journeys, carriers, intermediaries and multimodal hub managers would have to work under different regimes. Passengers would be subject to multiple rules and find it difficult to know and insist on their rights. Moreover, several regimes might apply to transport contracts for multimodal journeys between Member States. For these reasons, national rules, even assuming that they gave passengers a high level of protection, would not achieve essential Union objectives and would even frustrate their achievement. In sum, this initiative seeks to address a gap in EU legislation which, if left at national level, would create the risk of distortions or potential negative spill-over effects.

## Proportionality

As detailed in chapter 7 of the Impact Assessment accompanying this proposal, none of the policy options goes beyond what is necessary to reach the overall policy objectives.

#### • Choice of the instrument

The impact assessment evidenced that regulatory measures are necessary to achieve the objective of the initiative, whose main objective is to ensure an adequate level of protection of passenger in the context of multimodal journeys. The rules established by the present proposal should be applied in a uniform and effective manner across the European Union. A Regulation that is directly applicable appears to be the most appropriate instrument to ensure a coherent implementation of the measures envisaged in all EU Member States, while reducing the risk of distortion within the single market, which could result from differences into how EU Member States transpose the requirements into national law. A Regulation as a legal instrument has already been chosen by the EU in order to protect the rights of passengers travelling respectively by air, railways, bus and coach, and sea and inland waterways.

# 3. RESULTS OF EX-POST EVALUATIONS, STAKEHOLDER CONSULTATIONS AND IMPACT ASSESSMENTS

## • Ex-post evaluations/fitness checks of existing legislation

Since no legislation currently exists, no ex-post evaluation or fitness check was carried out.

#### Stakeholder consultations

For the impact assessment, stakeholders' input was gathered by various means, including an open public consultation (OPC) and targeted consultations by an external contractor through online questionnaires, interviews and workshops. Both qualitative input (opinions, views, suggestions) and quantitative information (data, statistics) were sought.

The stakeholders included participants from the industry, groups representing passengers/consumers, persons with disabilities and persons with reduced mobility (PRMs),

-

See footnote 2 above.

and public authorities, i.e. those affected by the policy, those who apply it and those with a vested interest in it.<sup>14</sup>

## Problem aspects

"Passengers are not protected by existing rights when switching between transport modes": 89 out of 145 respondents to the targeted questionnaire strongly or somewhat agree that the problem is important, while 26 respondents fully or somewhat disagree. The latter are mainly carriers (for all modes) and their umbrella organisations, who during the interviews and the dedicated workshop argued that it is too early to regulate the sector, as a legislative framework might disrupt an underdeveloped but emerging industry. In the OPC, 70% (117 out of 166) of respondents agreed somewhat/fully that the problem is important. In reply to the call for evidence, one passenger organisation claimed that the certainty of reaching the destination must be always guaranteed, even in the case of serious delays and independently from the travel mode(s) used.

"Persons with disabilities or reduced mobility (PRM) are not provided with assistance when switching between transport modes": 50% (69 out of 138) of respondents agree that this is an important problem, while 16% (22 out of 138) strongly or somewhat disagree, 14% (19 out of 138) are neutral and 20% (28 out of 138) expressed no opinion. In the OPC, 57% (95 out of 167) of respondents consider this to be an important problem.

## Policy objectives

71% (101 out of 143) of respondents to the targeted questionnaire agree with the objective of proposing an adequate framework of rules for the protection of passengers who experience travel disruptions when changing from one transport mode to another, while 11% (16 out of 143) disagree with this objective. In the OPC, 63% (104 out of 165) of respondents assess the objective's relevance as high or medium-high, while 16% (26 out of 165) believe it is of low/medium-low relevance.

## Policy measures

Most consumer organisations replying to the targeted questionnaire expressed support for a legislative measure to **ensure that carriers and ticket vendors provide additional information to multimodal passengers**. Airlines and their representative organisations indicated that they could not be legally bound to provide information that they often do not have (connecting times and connecting journeys) and pointed out that providing real-time information on journey disruption is unfeasible if passengers booked via a ticket vendor. Umbrella organisations explained that this measure would increase operational costs mainly due to the necessary updates of the IT systems supporting booking arrangements and to additional resources needed to identify the relevant information. Ticket vendors were less supportive of the provision of real-time information and claimed that such a solution would only be feasible if carriers were obliged to share information with intermediaries.

As to the **introduction of a right for passengers to be reimbursed or re-routed in case of a travel disruption**, most passenger organisations replying to the targeted survey stated that the legislative measure would bring a very strong or strong contribution, while a slight majority of carriers for rail, waterborne, bus and coach, believe that the legislative measure would have

\_

For more information on the stakeholder consultation, see the Synopsis report in Annex 2 of the Impact Assessment SWD(2023)386.

a moderate or slight contribution. Nearly all intermediaries replying to the targeted survey agreed that the proposed measure would have a strong contribution, if implemented as a legislative measure, and provided that there are B2B agreements with respective liabilities. As to the introduction of a **single point of contact for PRMs at multimodal passenger hubs**, one PRM representative underlined during the dedicated workshop that, as PRMs require different types of assistance, there would be many issues to consider in defining the role/scope of the SPC and expressed concerns if the measure was soft law. One interviewed PRM organisation remarked that the SPC would be of considerable help to PRMs to obtain assistance and relevant information. 10 out of 17 airlines stated that the measure would bring a (very) strong contribution. During the workshop, airlines and their umbrella organisation expressed their support for it, while stressing the importance of establishing clear channels of communication and defining the scope of the services to be provided. One interviewed organisation of airlines argued that the SPC should be the terminal operator, as in the case of air transport. 6 out of the 11 infrastructure and terminal managers replying to the targeted survey believe that the measure would (very) strongly contribute towards the policy objective.

## Collection and use of expertise

## Exploratory study

The Commission carried out an exploratory study on passenger rights in the multimodal context that was published in 2019. The results of this study were taken into account, in particular with regard to the identification of the legal gaps and the size of the multimodal passenger market. The study observes that the mode-oriented approach of the five existing passenger rights Regulations can potentially lead to legal gaps and, overall, an insufficient coverage of passenger rights in a multimodal context. The availability of data also constitutes a limitation in terms of both the scope and the detail of the analysis.

## Impact assessment

The impact assessment has been based on research and analyses done by the Commission. The Commission also contracted an external, independent consultant to support the impact assessment in specific tasks, i.e. the assessment of the costs and benefits as well as the open public consultation and targeted consultation (by means of interviews and workshops). The external support study will be published alongside this proposal.

#### • Impact assessment

An impact assessment report was submitted to the Regulatory Scrutiny Board (RSB) on 7 June 2023. The RSB subsequently issued a positive opinion with reservations on 7 July 2023. In particular, it identified a number of shortcomings in the report pertaining to the description of the scale of the problem and the magnitude of its consequences; the description of the benefits of the options and the assessment of the proportionality of the preferred option; as well as the outline of the administrative and adjustment costs as part of the One In, One Out assessment<sup>16</sup>. These shortcomings were all addressed in the final impact assessment report which is submitted together with this proposal, in particular by an improved description of the size of the multimodal market and the current market failure in mitigating the problem; the

SWD(2023)386, Annex 1.

https://op.europa.eu/en/publication-detail/-/publication/f176da6f-d9ca-11e9-9c4e-01aa75ed71a1.

magnitude of the consequences of the problem in terms of the lack of information and assistance during multimodal journeys; the benefits in terms of hassle costs savings for consumers, carriers, intermediaries and national public authorities; the comparison of the options, including with regard to proportionality; and the classification of one-off costs for implementing real-time information provisions and integrating the communication systems to multimodal passengers as adjustment costs rather than administrative costs.

The policy measures included in the three policy options addressing problem 2 of this impact assessment ("Insufficient protection of passengers during multimodal journeys") are differentiated between the three ticket categories (A-B-C) presented above. The table below contains an overview of the different policy options, while taking into account these different ticket categories.

	Category A	Category B	Category C
	(Single contract)	(Separate tickets combined and	(Separate tickets at
		sold by ticket vendor)	passenger's own initiative)
PO 2.1	Pre-purchase information	Pre-purchase information	Pre-purchase information
	(PM B.1)	(PMB.1)	(PMB.1)
	Basic set of passenger rights	Complaint-handling by industry	Complaint-handling by
	(PM B.3a)	and NEBs	industry and NEBs
	Complaint-handling by industry	(PM B.7 & B.8)	(PM B.7 & B.8)
	and NEBs		
	(PM B.7 & B.8)		
PO 2.2	Pre-purchase information	Pre-purchase information	Pre-purchase information
	(PM B.1)	(PM B.1)	(PM B.1)
	Real-time information	Real-time information	Single Point of Contact
	(PM B.2)	(PM B.2)	(PM B.5 & B.6)
	Basic set of passenger rights	Information on separate nature of	Complaint-handling by
	(PM B.3a)	tickets	industry and NEBs
	Single Point of Contact	(PM B.4a)	(PM B.7 & B.8)
	(PM B.5 & B.6)	Single Point of Contact	
	Complaint-handling by industry	(PM B.5 & B.6)	
	and NEBs	Complaint-handling by industry	
	(PM B.7 & B.8)	and NEBs	
		(PM B.7 & B.8)	
PO 2.3	Pre-purchase information	Pre-purchase information	Pre-purchase information
	(PM B.1)	(PM B.1)	(PM B.1)
	Real-time information	Real-time information	Single Point of Contact
	(PM B.2)	(PM B.2)	(PM B.5 & B.6)
	Enhanced set of passenger	Information on separate nature of	Complaint-handling by
	rights	tickets	industry and NEBs
	(PM B.3b)	(PM B.4b)	(PM B.7 & B.8)
	Single Point of Contact	Single Point of Contact	
	(PM B.5 & B.6)	(PM B.5 & B.6)	
	Complaint-handling by industry	Complaint-handling by industry	
	and NEBs	and NEBs (PM B.7 & B.8)	
	(PM B.7 & B.8)		

The impact assessment concluded that policy option PO 2.2 was the preferred option. The impact assessment found that it is the most efficient and effective policy option, as it ensures a fair balance between the effectiveness of the intervention in achieving the objective of increasing the protection of passengers (including PRM) in the context of multimodal journeys on the one hand, and efficiency in terms of costs for industry, on the other, given that this market is under development. PO 2.2 was found to also be the most proportionate measure and the most coherent in terms of internal and external coherence as well.

#### Sustainable Development Goals

The initiative contributes towards **Sustainable Development Goals** (SDG) #10 Reduce inequality within and among countries (regarding passengers with disabilities and reduced mobility), #13 Climate action (enhanced passenger rights may incentivise people to use more public transport instead of private cars), and #16 Provide access to justice for all and build effective, accountable and inclusive institutions (enhanced and accessible means of redress for passengers and better tools for enforcement for the administration).

## Climate consistency check

The proposal is consistent with climate neutrality objective set out in the European Climate Law and the Union 2030 and 2050 targets and contributes to achieve SDG #13 Climate action as mentioned above.

## Regulatory fitness and simplification

This initiative is part of the Commission Work Programme 2023 under Annex II (REFIT initiatives), under the heading 'An Economy that Works for People'. It has an important REFIT dimension in terms of the simplification of the rules for passengers travelling in a multimodal context.

In particular, a strong simplification potential resides in the improved information on the type of ticket that passengers are travelling with during a multimodal journey. Currently, they would not have easy access to such information, if at all available. Multimodal passengers would also have improved redress tools by means of complaint-handling procedures on the level of both industry and national authorities, making it easier for them to assert their rights. Together with increased rights in the event of missed connections between transport services (reimbursement or re-routing), this corresponds to a potential for cost savings for passengers. For PRM in particular, there is also an important efficiency potential where they would not only have a right to free assistance when travelling with a single multimodal contract, but would also have the possibility to receive a coordinated assistance when transferring at multimodal hubs (Single Points of Contact), where they currently would have to submit several assistance requests to carriers and terminal operators.

The proposal would necessarily imply a regulatory burden given that there are currently no rules protecting passengers undertaking multimodal journeys. However, there is a significant mitigating factor in this context, especially for transport operators, where they already apply most measures in the context of travels within a single mode of transport (e.g. reimbursement or re-routing, care, PRM assistance) – only the part of the journey when the passenger is switching between transport modes was not taken into consideration until now. In addition, carriers and intermediaries that qualify as SMEs would be exempted from the proposed rules on the provision of real-time information (Articles 5 and 6 of the proposal.)

The impact on EU competitiveness or international trade is expected to be largely neutral.

The proposal takes the digital environment into account, in particular with regard to the provision of real-time information on e.g. disruptions and delays - which are ideally also delivered by digital communication means — and the suggested rules on complaint-handling, where both companies and national enforcement bodies need to ensure that passengers can submit complaints both offline and online. The proposal also considers the increasing presence of online travel agents who would be willing to offer a combination of transport services to prospective passengers. This is not only reflected in terms of rules on the sharing of travel information, but also with regard to the liability of intermediaries selling a combination of ticket for different transport services.

## • Fundamental rights

Article 38 of the Charter of Fundamental Rights requires that Union policies ensure a high level of consumer protection. Article 26 of the Charter calls for the integration of persons with disabilities and requires Member States to take measures to ensure their independence, social and occupational integration, and participation in the life of the community. Strengthening the

rights of passengers travelling in a multimodal context in the EU will further raise the overall high level of consumer protection.

The obligation to share passengers' personal data has an impact on the right to protection of personal data under Article 8 of the Charter. This obligation is necessary and proportionate in light of the objective of ensuring that consumers are effectively protected in accordance with Article 38 of the Charter. In addition, to further mitigate the effects on data protection, data protection safeguards are implemented, namely purpose limitation and a clear retention period. Further provisions of the proposal concerning the use of passengers' details contribute to respect the right to protection of personal data under Article 8 of the Charter by clearly laying down which personal data can be used for specified purposes (for informing, reimbursing and rerouting the passenger as well as for safety and security reasons) and under which conditions.

## 4. **BUDGETARY IMPLICATIONS**

The proposal has no implications for the EU budget.

#### 5. OTHER ELEMENTS

## • Implementation plans and monitoring, evaluation and reporting arrangements

The Commission will monitor the implementation and effectiveness of this initiative on the basis of progress indicators mentioned in chapter 9 of the Impact Assessment. Five years after the entry into force of the proposed legislation, the Commission will evaluate whether its objectives were reached.

In its Communication on 'Long-term competitiveness of the EU: looking beyond 2030'<sup>17</sup>, the Commission committed to rationalise and simplify reporting requirements without undermining the related policy objectives. This proposal is in line with those aims. Although it creates a new obligation for carriers offering single multimodal contracts to publish every two years a report about the implementation of their service quality standards with key performance indicators related to passenger rights (e.g. missed connections, complaint handling, cooperation with the representatives of persons with disabilities), these reports are absolutely necessary for monitoring and enforcing passenger rights. Carriers indicated in the stakeholder consultation that they already collect the data in question, and the frequency of reporting (only every two years) was chosen to keep the costs of implementing these measures low. In addition of being a valuable source of information for NEBs with limited resources to carry out their monitoring and enforcement tasks, these reports will allow passengers to take informed decisions about which carrier to choose based on the quality of their performance and could also encourage competition between carriers based on the quality of their performance. In the same vein, the obligation for a biennial reporting by NEBs on actions taken to ensure the application of the Regulation and relevant statistics on e.g. complaints would allow the Commission in its task to verify the implementation of the Regulation.

<sup>17</sup> COM(2023)168.

## Detailed explanation of the specific provisions of the proposal

The structure of the Regulation is inspired by the existing Regulations on passenger rights, and in particular the most recent Regulation (EU) 2021/782 on rail passengers' rights and obligations. It will consist of the following main chapters:

#### Chapter I: General provisions

Chapter I contains the general provisions of the Regulation specifying the subject matter and objectives, the scope and the definitions. In addition, the complementary nature of the proposal to the existing Union law on passenger rights is outlined.

## Chapter II: Transport contracts and information

Chapter II contains provisions on transport contracts and information to passengers by carriers, intermediaries and multimodal hub managers. It describes the information to be given to passengers before and during their journey (in real-time), and the modalities for exchange and cooperation on the matter between different types of undertakings involved.

## Chapter III: Liability in the event of missed connections

Chapter III contains provisions on the assistance of passengers (reimbursement, re-routing, care) having a single multimodal contract in the event of a missed connection of a subsequent transport service. In addition, it spells out the reimbursement process where such contract was acquired with an intermediary. It also clarifies the liability of carriers and intermediaries offering combined multimodal tickets. Finally, it introduces a common form for reimbursement and compensation requests.

## Chapter IV: Persons with disabilities and reduced mobility

Chapter IV outlines rules for the protection and assistance of persons with disabilities and reduced mobility (PRM) in the context of multimodal travel. Next to detailing the right to transport and the assistance of PRM having a single multimodal contract, it introduces the establishment of Single Points of Contact at multimodal passenger hubs. These hubs correspond to the multimodal passenger hubs that have to be developed in every urban node of the TEN-T network by 2030 under the proposal for a Regulation on Union guidelines for the development of the trans-European transport network.<sup>18</sup>

## Chapter V: Service quality and complaints

Chapter V contains rules on service quality and on the handling of complaints by carriers, intermediaries and multimodal hub managers.

## Chapter VI: Information and enforcement

Chapter VI contains provisions on information to passengers on their rights as well as the enforcement of the Regulation. With regard to the latter, it includes rules on the designation of

\_

Proposal for a REGULATION OF THE EUROPEAN PARLIAMENT AND OF THE COUNCIL on Union guidelines for the development of the trans-European transport network, amending Regulation (EU) 2021/1153 and Regulation (EU) No 913/2010 and repealing Regulation (EU) 1315/2013, COM/2021/812 final of 14.12.2021 - 2021/0420(COD).

a national enforcement body, the risk-based approach to the monitoring of compliance with passenger rights, the sharing of information by the relevant undertakings with national enforcement bodies and cooperation between Member States and the Commission.

# Chapter VII: Final provisions

Chapter VII contains provisions with regard to reporting obligations and the committee procedure.

#### Annexes

Annex I sets out the list of urban nodes where multimodal passenger hubs and the corresponding Single Points of Contact need to be established and operated.

Annex II contains the minimum service quality standards that carriers offering single multimodal contracts and multimodal hub managers need to uphold.

## Proposal for a

## REGULATION OF THE EUROPEAN PARLIAMENT AND OF THE COUNCIL

## on passenger rights in the context of multimodal journeys

(Text with EEA relevance)

## THE EUROPEAN PARLIAMENT AND THE COUNCIL OF THE EUROPEAN UNION,

Having regard to the Treaty on the Functioning of the European Union, and in particular Article 91(1) and Article 100(2) thereof,

Having regard to the proposal from the European Commission,

After transmission of the draft legislative act to the national parliaments,

Having regard to the opinion of the European Economic and Social Committee<sup>1</sup>

Having regard to the opinion of the Committee of the Regions<sup>2</sup>

Acting in accordance with the ordinary legislative procedure,

#### Whereas:

- (1) The Union legislation on passenger rights<sup>3</sup> has significantly improved the protection of the rights of passengers travelling by air, sea and inland waterways, bus and coach, and rail when their travel plans are disrupted by long delays and cancellations.
- (2) The implementation of the rights established and protected under those Regulations has however revealed shortcomings for passengers who perform or intend to perform a journey involving a combination of transport modes, thereby preventing the full potential of passenger rights to be realised.

OJ C [...], [...], p. [...].

-

OJ C [...], [...], p. [...].

Regulation (EC) No 261/2004 of the European Parliament and of the Council of 11 February 2004 establishing common rules on compensation and assistance to passengers in the event of denied boarding and of cancellation or long delay of flights, and repealing Regulation (EEC) No 295/91, OJ L 46, 17.2.2004, p. 1, ELI: http://data.europa.eu/eli/reg/2004/261/oj; Regulation (EC) No 1107/2006 of the European Parliament and of the Council of 5 July 2006 concerning the rights of disabled persons and persons with reduced mobility when travelling by air, OJ L 204, 26.7.2006, p. 1, ELI: http://data.europa.eu/eli/reg/2006/1107/oj; Regulation (EU) 2021/782 of the European Parliament and of the Council of 29 April 2021 on rail passengers' rights and obligations (recast), OJ L 172, 17.5.2021, p. 1, ELI: http://data.europa.eu/eli/reg/2021/782/oj; Regulation (EU) No 1177/2010 of the European Parliament and of the Council of 24 November 2010 concerning the rights of passengers when travelling by sea and inland waterway and amending Regulation (EC) No 2006/2004, OJ L 334, 17.12.2010, p. 1, ELI: http://data.europa.eu/eli/reg/2010/1177/oj; Regulation (EU) No 181/2011 of the European Parliament and of the Council of 16 February 2011 concerning the rights of passengers in bus and coach transport and amending Regulation (EC) No 2006/2004, OJ L 55, 28.2.2011, p. 1, ELI: http://data.europa.eu/eli/reg/2011/181/oj.

- (3) In the framework of the common transport policy, it is important to safeguard the rights of passengers switching modes of transport in order to assist the development of multimodal travel and improve the choice for passengers in terms of travel options.
- (4) The Union standards of protection set by Regulations (EC) No 261/2004, (EC) No 1107/2006, (EU) 2021/782, (EU) No 1177/2010 and (EU) No 181/2011 for travelling using one mode of transport should therefore be extended to cover the rights of passengers switching between modes of transport and to ensure that undertakings which are involved in a multimodal journey operate under harmonised rules.
- (5) Legal protection should be provided for passengers who perform multimodal journeys where there is no specific sectoral Union legislation on the matter.
- (6) The rights for passengers who perform multimodal journeys established in this Regulation should complement the protection that those same passengers already enjoy under the Union legislation on passenger rights. Therefore, this Regulation should be without prejudice to the rights and obligations under that legislation.
- (7) This Regulation should not apply to single multimodal contracts where they are part of a package and a right to reimbursement arises under Directive (EU) 2015/2302 of the European Parliament and Council<sup>4</sup>. In view of the protection to passengers foreseen under that Directive, this Regulation should also not apply to combined multimodal tickets where they are combined by an organiser as part of a package.
- (8) Member States should ensure that discrimination on the basis of the nationality of the passenger or the place of establishment within the Union of the carrier or intermediary is prohibited when carriers or intermediaries offer contract conditions and tariffs for multimodal journeys to the general public. Social tariffs should not be prohibited, provided that such measures are proportionate and independent of the nationality of the passenger concerned. In light of the development of online platforms selling passenger transport tickets, Member States should pay special attention to ensuring that no discrimination on the basis of nationality of the passenger or the place of establishment within the Union of the carrier or intermediary occurs during the process of accessing online interfaces or purchasing tickets. Furthermore, regardless of how a certain type of a ticket is purchased, the level of protection of the passenger should be the same.
- (9) Member States should ensure that carriers and intermediaries offering transport contracts for the purpose of a multimodal journey inform the passenger of the type of ticket or tickets associated to that journey and their corresponding rights, in particular with regard to missed connections.
- (10) Access to travel information, including real-time data, makes multimodal travel easier and provides passengers with a wider range of journey possibilities. In this regard, Commission Delegated Regulation (EU) 2017/1926<sup>5</sup> sets out the accessibility of such travel and traffic information data to National Access Points, in order to ensure that

-

Directive (EU) 2015/2302 of the European Parliament and of the Council of 25 November 2015 on package travel and linked travel arrangements, amending Regulation (EC) No 2006/2004 and Directive 2011/83/EU of the European Parliament and of the Council and repealing Council Directive 90/314/EEC (OJ L 326, 11.12.2015, p. 1, ELI: http://data.europa.eu/eli/dir/2015/2302/oj).

Commission Delegated Regulation (EU) 2017/1926 of 31 May 2017 supplementing Directive 2010/40/EU of the European Parliament and of the Council with regard to the provision of EU-wide multimodal travel information services (OJ L 272, 21.10.2017, p. 1, ELI: http://data.europa.eu/eli/reg\_del/2017/1926/oj).

ticket vendors can inform passengers before and during their journey. In order to avoid any unnecessary burden for carriers offering single multimodal contracts, they should be able to comply with provisions under this Regulation to provide travel information to other carriers and intermediaries selling their services to the extent where they provide this data to such National Access Points.

- (11) Carriers and intermediaries that are small and medium-sized enterprises (SMEs) fulfilling the criteria laid down in Annex I to Commission Regulation (EU) No 651/2014<sup>6</sup> often have limited resources, which may restrict their access to information, notably in the context of new technology. Therefore, such carriers and intermediaries should be exempted from the requirements on the provision of real-time information to multimodal passengers.
- (12) Passengers having a single multimodal contract and who miss a connection due to a delay or cancellation of a preceding service under that contract should be able either to obtain reimbursement of their tickets or to obtain re-routing under satisfactory conditions, and they should be adequately cared for while awaiting a later connection.
- (13) Where a single multimodal contract is booked via an intermediary, the intermediary and the contracting carrier should inform the passenger about the reimbursement process. In particular, the contracting carrier should state publicly whether it cooperates with intermediaries for the processing of reimbursements, and if so with which it does so.
- (14) Correct information as regards a multimodal transport service is also essential when passengers buy tickets from intermediaries. Where intermediaries sell separate tickets for different transport services as a bundle in the form of a combined multimodal ticket, they should clearly inform the passenger that those tickets do not offer the same level of protection as single multimodal contracts and that those tickets have not been issued as single multimodal contracts by the carrier or carriers providing the service. Where intermediaries fail to comply with this requirement, their liability should include the reimbursement of the ticket amount and an additional compensation equivalent to 75% of that amount.
- (15) In light of the United Nations Convention on the Rights of Persons with Disabilities and in order to give persons with disabilities and persons with reduced mobility opportunities for multimodal travel comparable to those of other citizens, rules for non-discrimination and assistance during their multimodal journey should be established. In particular, special attention should be given to the provision of information to persons with disabilities and persons with reduced mobility concerning the accessibility of transport services, access conditions of vehicles and the facilities on board when switching modes. If information to persons with disabilities and reduced mobility is provided in accessible formats, it should be provided in accordance with the applicable legislation such as the accessibility requirements set out in Annex I to Directive (EU) 2019/882<sup>7</sup>. In light of the benefits in terms of security, convenience and accessibility, Member States should encourage the use of

.

Commission Regulation (EU) No 651/2014 of 17 June 2014 declaring certain categories of aid compatible with the internal market in application of Articles 107 and 108 of the Treaty (OJ L 187, 26.6.2014, p. 1, ELI: <a href="http://data.europa.eu/eli/reg/2014/651/oj">http://data.europa.eu/eli/reg/2014/651/oj</a>).

Directive (EU) 2019/882 of the European Parliament and of the Council of 17 April 2019 on the accessibility requirements for products and services (OJ L 151, 7.6.2019, p. 70), ELI: http://data.europa.eu/eli/dir/2019/882/oj.

- European digital identity wallets for identification and authentication in multimodal transport scenarios, particularly aiding vulnerable persons or persons with disabilities.
- (16) Carriers and multimodal hub managers should actively cooperate with organisations representing people with disabilities in order to improve the quality of accessibility of transport services. In order to facilitate access to multimodal passenger services for persons with disabilities and persons with reduced mobility, Member States, carriers and terminal managers should set up national Single Points of Contact to coordinate information and assistance at multimodal passenger hubs in certain major urban nodes.
- (17) In addition, Member States should have the possibility to require carriers and terminal managers to set up national Single Points of Contact to coordinate information and assistance at additional multimodal passenger hubs.
- (18) Carriers and terminal managers should define, manage and monitor service quality standards for multimodal passenger services. Carriers should also make information on their service quality performance publicly available.
- (19) This Regulation should not affect the rights of passengers to file a complaint with a national body or to seek legal redress through national procedures.
- (20) In order to maintain a high level of consumer protection in multimodal transport, Member States should be required to designate national enforcement bodies to monitor closely the application of this Regulation and to enforce it at national level. Those bodies should be able to take a variety of enforcement measures. Passengers should be able to complain to those bodies, or any other body designated by a Member State to this effect, about alleged infringements of the Regulation. It should be noted that Member States may choose to designate a national enforcement body that is also designated as the body responsible for the enforcement of other Union rules on passenger rights.
- (21) Processing of personal data should be carried out in accordance with Union law on the protection of personal data as laid down in Regulation (EU) 2016/679 of the European Parliament and of the Council.<sup>8</sup> Any processing of personal data must in particular take place in accordance with the requirements set out in Article 5(1) and Article 6(1) of that Regulation. It should also be noted that the obligations to provide information to passengers concerning their rights are without prejudice to the obligation of the controller to provide information to the data subject pursuant to Articles 12, 13 and 14 of Regulation (EU) 2016/679.
- (22) In order to ensure that a passenger receives travel information about a multimodal journey, both before and during that journey, the sharing of passenger contact details with the carrier could be necessary. The carrier may use these contact details exclusively for the purpose of fulfilling the information obligation under this Regulation and to fulfil the carrier's obligations under applicable Union law on safety and security. This personal data should not be processed for any other purposes and should be deleted within 72 hours after the completion of the contract of carriage unless further retention of the contact details is justified to fulfil obligations in respect of the passenger's right to re-routing, reimbursement or compensation.

-

Regulation (EU) 2016/679 of the European Parliament and of the Council of 27 April 2016 on the protection of natural persons with regard to the processing of personal data and on the free movement of such data, and repealing Directive 95/46/EC (General Data Protection Regulation) (OJ L 119, 4.5.2016, p. 1, ELI: <a href="http://data.europa.eu/eli/reg/2016/679/oj">http://data.europa.eu/eli/reg/2016/679/oj</a>).

- (23) Member States should lay down penalties applicable to infringements of this Regulation and ensure that these penalties are applied. The penalties should be effective, proportionate and dissuasive.
- (24) Since the objectives of this Regulation, namely the development of the Union's market for multimodal passenger transport and the establishment of passengers' rights in the context of multimodal journeys, cannot be sufficiently achieved by the Member States, and can therefore be better achieved at Union level, the Union may adopt measures, in accordance with the principle of subsidiarity as set out in Article 5 of the Treaty on European Union. In accordance with the principle of proportionality, as set out in that Article, this Regulation does not go beyond what is necessary in order to achieve those objectives.
- (25) In order to ensure uniform conditions for the implementation of this Regulation, implementing powers should be conferred on the Commission. Those powers should be exercised in accordance with Regulation (EU) No 182/2011 of the European Parliament and of the Council.<sup>9</sup> The examination procedure should be used for the adoption of the common forms for reimbursement and compensation requests.
- (26) This Regulation respects fundamental rights and observes the principles recognised in the Charter of Fundamental Rights of the European Union, in particular Articles 21, 26, 38 and 47 concerning, respectively, the prohibition of any form of discrimination, the integration of persons with disabilities, the ensuring of a high level of consumer protection, and the right to an effective remedy and to a fair trial. The Member States' courts must apply this Regulation in a manner consistent with these rights and principles,
- (27) The European Data Protection Supervisor was consulted in accordance with Article 42(1) of Regulation (EU) 2018/1725 of the European Parliament and of the Council and delivered an opinion on [],

HAVE ADOPTED THIS REGULATION:

#### **CHAPTER I**

#### **GENERAL PROVISIONS**

## Article 1

## **Subject matter**

This Regulation establishes rules applicable to multimodal transport as regards the following:

- (a) non-discrimination between passengers with regard to transport conditions and the provision of tickets;
- (b) information to passengers;
- (c) passengers' rights in the event of disruption, in particular in the context of a missed connection between different transport modes;

Regulation (EU) No 182/2011 of the European Parliament and of the Council of 16 February 2011 laying down the rules and general principles concerning mechanisms for control by the Member States of the Commission's exercise of implementing powers (OJ L 55, 28.2.2011, p. 13, ELI: http://data.europa.eu/eli/reg/2011/182/oj).

- (d) non-discrimination against, and assistance for, persons with disabilities and persons with reduced mobility;
- (e) the definition and monitoring of service quality standards;
- (f) handling of complaints;
- (g) general rules on enforcement;
- (h) penalties.

## Scope

- 1. This Regulation shall apply to multimodal journeys, of which all the transport services concerned fall under the scope of the Union legislation on passenger rights<sup>10</sup>, offered by carriers or intermediaries to passengers in the form of:
  - (a) a single multimodal contract,
  - (b) a combined multimodal ticket,
  - (c) separate multimodal tickets.
- 2. This Regulation shall apply to carriers, intermediaries and terminal managers. It shall also apply to multimodal hub managers operating Single Points of Contact at multimodal passenger hubs at the urban nodes listed in Annex I.
- 3. This Regulation is without prejudice to the following rules laid down by other Union legislation regulating other aspects of the protection of passengers, and complements that protection:
  - (a) Union legislation on passenger rights;
  - (b) Union legislation on package travel and linked travel arrangements<sup>11</sup>;
  - (c) Union legislation on consumer protection<sup>12</sup>.

<sup>10</sup> Regulation (EC) No 261/2004 of the European Parliament and of the Council of 11 February 2004 establishing common rules on compensation and assistance to passengers in the event of denied boarding and of cancellation or long delay of flights, and repealing Regulation (EEC) No 295/91, (OJ L 46, 17.2.2004, p. 1), ELI: http://data.europa.eu/eli/reg/2004/261/oj; Regulation (EC) No 1107/2006 of the European Parliament and of the Council of 5 July 2006 concerning the rights of disabled persons and persons with reduced mobility when travelling by air, (OJ L 204, 26.7.2006, p. 1, ELI: http://data.europa.eu/eli/reg/2006/1107/oj); Regulation (EU) 2021/782 of the European Parliament and of the Council of 29 April 2021 on rail passengers' rights and obligations (recast), (OJ L 172, 17.5.2021, p. 1, ELI: http://data.europa.eu/eli/reg/2021/782/oj); Regulation (EU) No 1177/2010 of the European Parliament and of the Council of 24 November 2010 concerning the rights of passengers when travelling by sea and inland waterway and amending Regulation (EC) No 2006/2004, (OJ L 334, 17.12.2010, p. 1, ELI: http://data.europa.eu/eli/reg/2010/1177/oj); Regulation (EU) No 181/2011 of the European Parliament and of the Council of 16 February 2011 concerning the rights of passengers in bus and coach transport and amending Regulation (EC) No 2006/2004, (OJ L 55, 28.2.2011, p. 1, ELI: http://data.europa.eu/eli/reg/2011/181/oj.)

Directive (EU) 2015/2302 of the European Parliament and of the Council of 25 November 2015 on package travel and linked travel arrangements, amending Regulation (EC) No 2006/2004 and Directive 2011/83/EU of the European Parliament and of the Council and repealing Council Directive 90/314/EEC (OJ L 326, 11.12.2015, p. 1, ELI: http://data.europa.eu/eli/dir/2015/2302/oj).

Regulation (EU) 2017/2394 of the European Parliament and of the Council of 12 December 2017 on cooperation between national authorities responsible for the enforcement of consumer protection laws

- 4. This Regulation shall not apply to combined multimodal tickets where they are combined by an organiser as part of a package under Directive (EU) 2015/2302<sup>13</sup>.
- 5. Paragraph 1, point (a) of Article 7 shall apply to passengers whose single multimodal contracts form part of a package, unless where a right to reimbursement arises under Directive (EU) 2015/2302<sup>14</sup>.

#### **Definitions**

For the purposes of this Regulation, the following definitions apply:

- (1) 'multimodal journey' means a journey of a passenger between a point of departure and a final destination covering at least two transport services and at least two modes of transport;
- (2) 'carrier' means a natural or legal person, other than an intermediary, offering transport services to the general public, including:
  - (a) operating air carriers as defined in point (b) of Article 2 of Regulation (EC) No 261/2004;
  - (b) railway undertakings as defined in point (1) of Article 3 of Regulation (EU) 2021/782;
  - (c) carriers as defined in point (d) of Article 3 of Regulation (EU) No 1177/2010;
  - (d) carriers as defined in point (e) of Article 3 of Regulation (EU) No 181/2011;
- (3) 'ticket vendor' means any natural or legal person, other than a carrier, acting on behalf of a carrier or a passenger for the conclusion of transport contracts;

and repealing Regulation (EC) No 2006/2004 (Text with EEA relevance) (OJ L 345, 27.12.2017, p. 1, ELI: http://data.europa.eu/eli/reg/2017/2394/oj,); Directive 2005/29/EC of the European Parliament and of the Council of 11 May 2005 concerning unfair business-to-consumer commercial practices in the internal market and amending Council Directive 84/450/EEC, Directives 97/7/EC, 98/27/EC and 2002/65/EC of the European Parliament and of the Council and Regulation (EC) No 2006/2004 of the European Parliament and of the Council ('Unfair Commercial Practices Directive') (Text with EEA relevance) (OJ L 149, 11.6.2005, p. 22, ELI: http://data.europa.eu/eli/dir/2005/29/oj); Directive 2013/11/EU of the European Parliament and of the Council of 21 May 2013 on alternative dispute resolution for consumer disputes and amending Regulation (EC) No 2006/2004 and Directive 2009/22/EC (Directive on consumer ADR) (OJ L 165, 18.6.2013, ELI: http://data.europa.eu/eli/dir/2013/11/oj); Directive (EU) 2020/1828 of the European Parliament and of the Council of 25 November 2020 on representative actions for the protection of the collective interests of consumers and repealing Directive 2009/22/EC (OJ L 409, 4.12.2020, p. 1 ELI: http://data.europa.eu/eli/dir/2020/1828/oj,); Council Directive 93/13/EEC of 5 April 1993 on terms contracts (OJ unfair in consumer L 95, 21.4.1993, p. http://data.europa.eu/eli/dir/1993/13/oj).

Directive (EU) 2015/2302 of the European Parliament and of the Council of 25 November 2015 on package travel and linked travel arrangements, amending Regulation (EC) No 2006/2004 and Directive 2011/83/EU of the European Parliament and of the Council and repealing Council Directive 90/314/EEC (OJ L 326, 11.12.2015, p. 1, ELI: http://data.europa.eu/eli/dir/2015/2302/oj).

Directive (EU) 2015/2302 of the European Parliament and of the Council of 25 November 2015 on package travel and linked travel arrangements, amending Regulation (EC) No 2006/2004 and Directive 2011/83/EU of the European Parliament and of the Council and repealing Council Directive 90/314/EEC (OJ L 326, 11.12.2015, p. 1, ELI: http://data.europa.eu/eli/dir/2015/2302/oj).

- (4) 'intermediary' means a ticket vendor or an organiser or retailer as defined in points (8) and (9) respectively of Article 3 of Directive (EU) 2015/2302 other than a carrier;
- (5) 'ticket' means valid evidence, regardless of its form, of the conclusion of a transport contract;
- (6) 'transport contract' means a contract of carriage between a carrier and a passenger for the provision of one or more transport services;
- (7) 'single multimodal contract' means a transport contract for a multimodal journey containing successive transport services operated by one or more carriers;
- (8) 'combined multimodal ticket' means a ticket or tickets for a multimodal journey representing separate transport contracts which are combined by a carrier or intermediary on its own initiative, and which are purchased by means of a single payment by the passenger;
- (9) 'separate multimodal tickets' mean tickets for a multimodal journey representing separate transport contracts which are offered together by a carrier or intermediary, and which are purchased by means of separate payments by the passenger;
- (10) 'transport service' means a passenger transport service that operates between terminals according to a timetable, including transport services offered for re-routing;
- (11) 'managing body of the airport' means a body as defined in point (f) of Regulation (EC) No 1107/2006;
- (12) 'airport' means an airport as defined in point (j) of Regulation (EC) No 1107/2006;
- (13) 'railway station manager' means a station manager as defined in point (3) of Article 3 of Regulation (EU) 2021/782;
- (14) 'railway station' means a station as defined in point (22) of Article 3 of Regulation (EU) 2021/782;
- (15) 'port terminal' means a terminal as defined in point (k) of Article 3 of Regulation (EU) No 1177/2010;
- (16) 'port terminal operator' means a terminal operator as defined in point (s) of Article 3 of Regulation (EU) No 1177/2010;
- (17) 'bus or coach terminal' means a terminal as defined in point (m) of Article 3 of Regulation (EU) No 181/2011;
- (18) 'bus terminal managing body' means an entity as defined in point (o) of Article 3 of Regulation (EU) No 181/2011;
- (19) 'terminal' means an airport, railway station, port terminal or bus or coach terminal;
- (20) 'terminal manager' means a managing body of the airport, a railway station manager, a port terminal operator or a bus terminal managing body
- 'multimodal passenger hub' means a connection point between at least two transport modes for passengers, where access to public transport and transfers between modes, including Park and Ride stations and active modes, are ensured and which act as an interface between urban nodes and longer-distance transport networks;
- (22) 'multimodal hub manager' means a terminal manager responsible for the management of a multimodal passenger hub;

- 'missed connection during a multimodal journey' means a situation where a passenger misses one or more transport services in the course of a multimodal journey, as a result of the delay or cancellation of one or more previous transport services, or of the departure of a transport service before the scheduled departure time;
- 'person with disabilities' and 'person with reduced mobility' mean any person who has a permanent or temporary physical, mental, intellectual or sensory impairment which, in interaction with various barriers, may hinder his or her full and effective use of transport on an equal basis with other passengers or whose mobility when using transport is reduced due to age;
- (25) 'small and medium-sized enterprises' or 'SMEs' means undertakings fulfilling the criteria laid down in Annex I to Commission Regulation (EU) No 651/2014;

#### **CHAPTER II**

#### TRANSPORT CONTRACTS AND INFORMATION

#### Article 4

## Non-discriminatory contract conditions and tariffs

- 1. Without prejudice to social tariffs, carriers or intermediaries shall offer contract conditions and tariffs for multimodal journeys to the general public without direct or indirect discrimination on the basis of the passenger's nationality or of the place of establishment within the Union of the carrier or intermediary.
- 2. Social tariffs shall be acceptable provided that they do not discriminate on the basis of nationality of the passenger.

#### Article 5

## **Travel information for passengers**

- 1. Carriers and intermediaries offering transport contracts on behalf of one or more carriers shall provide the passenger with information prior to purchase on whether the ticket or the tickets offered for a multimodal journey constitute a single multimodal contract, a combined multimodal ticket or separate multimodal tickets, as well as on the rights associated with the type of contract or ticket.
- 2. Prior to purchase, carriers and intermediaries offering transport contracts on behalf of one or more carriers for the purpose of a multimodal journey shall provide the passenger with general guidance on minimum connecting times between the different types of transport services offered in a multimodal journey.
- 3. Multimodal hub managers shall also provide general guidance on minimum connecting times between the different types of transport services that operate at the multimodal hub.
- 4. Carriers and intermediaries offering transport contracts on behalf of one or more carriers for the purpose of a multimodal journey shall provide the passenger, with the following information before the multimodal journey:
  - (a) general conditions applicable to the contract;

- (b) time schedules and conditions for the fastest trip for the multimodal journey;
- (c) time schedules and conditions for all available fares for the multimodal journey, highlighting the lowest fares;
- (d) disruptions and delays affecting the multimodal journey, planned and in real time;
- (e) procedures for the submission of complaints under Article 18.
- 5. Carriers offering single multimodal contracts and combined multimodal tickets and, where possible, intermediaries offering combined multimodal tickets shall provide the passenger during the journey with the following information:
  - (a) disruptions and delays (planned and in real time);
  - (b) main connecting transport services;
  - (c) security and safety issues occurring on board the transport service and at terminals.
- 6. The information referred to in paragraphs 1 to 5 shall be provided in the most appropriate format, including by using appropriate communication technologies. This information shall be provided in an accessible format.
- 7. The information to passengers shall be provided also by electronic means, where technically possible. Where information is provided by electronic means of communication, carriers and intermediaries shall ensure that the passenger can keep any written correspondence, including the date and time of such correspondence, on a durable medium. All means of communication shall enable the passenger to contact them quickly and to communicate effectively.
- 8. Where the passenger does not acquire a single multimodal contract directly from the carrier, but via an intermediary, this intermediary shall provide the contact details of the passenger and the booking details to the carriers concerned. The carrier may only use these contact details to the extent necessary to comply with its obligations under this Regulation to provide information, provision of care, reimbursement, re-routing and compensation, and to fulfil the carrier's obligations under applicable Union law on safety and security. The carrier shall delete the contact details within 72 hours after the completion of the contract of carriage unless further retention of the contact details is justified to fulfil obligations in respect of the passenger's right to rerouting, reimbursement or compensation.
- 9. Carriers and intermediaries which are SMEs shall be exempted from the provisions on real-time information under this Article.

## Access to travel information for carriers and intermediaries

- 1. Without prejudice to Article 10 (1) of Regulation (EU) 2021/782, carriers offering single multimodal contracts shall provide other carriers and intermediaries which sell their contracts with access to the travel information set out in Article 5(2) to (5).
- 2. Carriers offering single multimodal contracts shall distribute this information and grant access in a non-discriminatory manner and without undue delay. A one-off request shall be sufficient to have continuous access to information. The carrier obliged to make available information in accordance with paragraph 1 may request

the conclusion of a contract or other arrangement on whose basis information is distributed or access is granted. The terms and conditions of any contract or arrangement for the use of the information shall not unnecessarily restrict possibilities for its reuse. The terms and conditions shall not be used to restrict competition. Carriers offering single multimodal contracts may require from other carriers and intermediaries a fair, reasonable and proportionate financial compensation for the costs incurred in providing the access.

- 3. Information shall be distributed, and access shall be provided by appropriate technical means, such as application programming interfaces. It shall be ensured that these application programming interfaces are in compliance with the specifications laid down in the implementing acts adopted under Directive (EU) 2016/797.
- 4. Where the information covered by paragraph 1 is provided in accordance with other Union legal acts, in particular delegated acts adopted under Directive 2010/40/EU of the European Parliament and of the Council<sup>15</sup>, the corresponding obligations under this Article are equally fulfilled.

#### **CHAPTER III**

#### MISSED CONNECTIONS

#### Article 7

## Reimbursement and re-routing

- 1. Where a missed connection of a subsequent transport service during a multimodal journey, concluded under a single multimodal contract, occurs or is reasonably to be expected to occur due to a delay or cancellation of a preceding transport service under that same contract, the contracting carrier shall immediately offer the passenger the choice between one of the following options:
  - (a) reimbursement of the full cost of the ticket, under the conditions by which it was paid for the part or parts of his or her journey not made and for the part or parts already made if the journey is no longer serving any purpose in relation to the passenger's original travel plan, together with, when relevant, a return service to the first point of departure at the earliest opportunity;
  - (b) continuation or re-routing, under comparable transport conditions, to the final destination at the earliest opportunity;
  - (c) continuation or re-routing, under comparable transport conditions, to the final destination at a later date at the passenger's convenience.

The contracting carrier shall make the arrangements necessary for the option chosen by the passenger.

2. Where, for the purposes of points (b) and (c) of paragraph 1, comparable re-routing is operated by the same carrier or another carrier is commissioned to perform the rerouting, this shall not generate additional costs to the passenger. That requirement

Directive 2010/40/EU of the European Parliament and of the Council of 7 July 2010 on the framework for the deployment of Intelligent Transport Systems in the field of road transport and for interfaces with other modes of transport (OJ L 207, 6.8.2010, p. 1, ELI: <a href="http://data.europa.eu/eli/dir/2010/40/oj">http://data.europa.eu/eli/dir/2010/40/oj</a>).

also applies where the re-routing involves the use of transport of a higher service class and alternative modes of transport. Carriers shall make reasonable efforts to avoid additional connections and to ensure that delay in the total travel time is as short as possible.

- 3. Re-routing transport service providers shall provide persons with disabilities and persons with reduced mobility with a level of assistance and accessibility comparable to the missed transport service when offering an alternative service. Re-routing transport service providers shall pay particular attention to provide persons with disabilities and persons with reduced mobility with alternative services which are appropriate to their needs, and which differ from those offered to other passengers.
- 4. The reimbursement referred to in point (a) of paragraph 1 shall be paid within 14 days after the receipt of the request. Member States may require contracting carriers to accept such requests by particular means of communication, provided that the requirement does not create discriminatory effects. The reimbursement may take the form of money, vouchers or the provision of other services provided that the terms of those vouchers and services are sufficiently flexible, in particular regarding the validity period and destination, and that the passenger expressly agrees to accept those vouchers and services. The reimbursement of the ticket price shall not be reduced by financial transaction costs such as fees, telephone costs or stamps.

#### Article 8

# Reimbursement when the single multimodal contract was booked through an intermediary

- 1. Where the passenger has bought the single multimodal contract through an intermediary, the contracting carrier may make the reimbursement referred to in Article 7(1) point (a) through that intermediary in accordance with this Article.
- 2. The intermediary and the contracting carrier shall inform the passenger about the reimbursement process as provided for in this Article in a clear, comprehensible and easily accessible manner at the time of booking and on the booking confirmation.
- 3. Reimbursement through the intermediary shall be free of charge for passengers and all other parties concerned.
- 4. The carrier shall state publicly, in a clear, comprehensible and easily accessible manner, whether it agrees to process reimbursements through intermediaries, and with which intermediaries it accepts to do so.
- 5. The following shall apply in the case of reimbursement through intermediaries which have paid the contracting carrier for the single multimodal contract from their own accounts:
  - (a) the contracting carrier shall reimburse the intermediary within seven days, in one transaction through the same payment method which was used at the time of booking, and linking the payment to the original booking reference. The seven-day period shall start on the date of the passenger's choice of a reimbursement in accordance with Article 7(1), point (a), of this Regulation. The intermediary shall reimburse the passenger via the original payment method, at the latest within a further seven days, and inform the passenger and the carrier thereof:

(b) if the passenger does not receive the reimbursement within 14 days as of the date of choosing a reimbursement in accordance with Article 7(1), point (a)of this Regulation, the contracting carrier shall contact the passenger at the latest on the day following the expiry of the 14-day period in order to receive the payment details for the reimbursement. Upon receipt of these payment details, the contracting carrier shall reimburse the passenger within 14 days and inform the passenger and the intermediary thereof.

#### Article 9

#### **Assistance**

- 1. In the case of a missed connection of a subsequent transport service during a multimodal journey concluded under a single multimodal contract, which is due to a delay or cancellation of a preceding transport service under that same contract, the contracting carrier shall offer the passengers the following free of charge:
  - (a) meals and refreshments in reasonable relation to the waiting time, if they are available on the transport service or in the terminal, or can reasonably be supplied, taking into account criteria such as the distance from the supplier, the time required for delivery and the cost;
  - (b) hotel or other accommodation, and transport between the terminal and place of accommodation, in cases where a stay of one or more nights or an additional stay becomes necessary, where and when physically possible. In cases where such a stay becomes necessary due to the circumstances referred to in Article 19(10) of Regulation (EU) 2021/782, the carrier may limit the duration of accommodation to a maximum of three nights. The access requirements of persons with disabilities and persons with reduced mobility and the needs of assistance dogs shall be taken into account, whenever possible.
- 2. In applying paragraph 1, the operating carrier shall pay particular attention to the needs of persons with disabilities and persons with reduced mobility, as well as to those of any accompanying persons and assistance dogs.

#### Article 10

#### Liability for combined multimodal tickets

- 1. A carrier or intermediary which sells a combined multimodal ticket shall be liable to reimburse the total amount paid for that ticket and, moreover, to pay compensation equivalent to 75 % of that amount if the passenger misses one or more connections. The right to reimbursement or to compensation shall be without prejudice to applicable national law granting passengers further compensation for damage.
- 2. The liability set out in paragraph 1 shall not apply if it is explicitly and clearly mentioned on the tickets, or on another document or electronically in such a manner that allows the passenger to reproduce the information for future reference, that the combined multimodal ticket consists of separate transport contracts with no right under this Regulation to reimbursement, re-routing, assistance or compensation in case of missed connections, and if the passenger was clearly informed of this prior to the purchase. The burden of proof that the passenger was provided with the information shall lie with the carrier or intermediary that sold the combined multimodal ticket.

- The burden of proof that the passenger was provided with the information shall lie with the carrier or intermediary which sold the combined multimodal ticket.
- 3. The carriers or intermediaries which sold the combined multimodal ticket shall be responsible for handling requests and possible complaints of the passenger under paragraph 1.
- 4. The reimbursement and the compensation referred to in paragraph 1 shall be paid within 14 days after the receipt of the request.

## Common form for reimbursement and compensation requests

- 1. The Commission shall adopt an implementing act establishing a common form for compensation and reimbursement requests under this Regulation. That common form shall be established in accessible formats. That implementing act shall be adopted in accordance with the examination procedure referred to in Article 26(2).
- 2. Passenger shall have the right to submit their requests using the common form referred to in paragraph 1. Carriers and intermediaries shall not reject a request for reimbursement or compensation solely on the grounds that the passenger has not used that form. If a request is not sufficiently precise, the carriers and intermediaries shall ask the passenger to clarify the request and shall assist the passenger in doing so.
- 3. Carriers and intermediaries shall provide details on their website such as an e-mail address to which requests under paragraph 1 may be sent by electronic means. That requirement shall not apply where other electronic means of communication allowing passengers to request reimbursement or compensation are available, such as a form on a website or mobile applications, provided that such means offer the choice and information set out in the common form and are also available in an official language of the Union and in the language internationally accepted in this field. When using such means, passengers shall not be prevented from providing information in any of the languages of the Union.
- 4. The Commission shall make the common form available in all Union languages on its website. The body or bodies designated by Member States in accordance with Articles 20(1) and 21(1) shall ensure that passengers have access to the common form.

#### **CHAPTER IV**

#### PERSONS WITH DISABILITIES AND PERSONS WITH REDUCED MOBILITY

## Article 12

### Right to transport

1. Carriers offering single multimodal contracts and multimodal hub managers shall establish non-discriminatory access rules for the transport of persons with disabilities, and the transport of persons with reduced mobility. Those rules shall comply with the relevant provisions on the limitation of transport of persons with disabilities and persons with reduced mobility in the Union law on passenger rights.

- 2. Reservations and tickets for multimodal journeys, whether in the form of a single multimodal contract, a combined multimodal ticket or separate multimodal tickets, shall be offered to persons with disabilities and persons with reduced mobility at no additional cost. A carrier or intermediary may not refuse to accept a reservation from, or to issue a ticket to, a person with disabilities or a person with reduced mobility or require that such person be accompanied by another person, unless this is strictly necessary in order to comply with the access rules referred to in paragraph 1.
- 3. The access rules referred to in paragraph 1 shall be established with the active involvement of representative organisations for persons with disabilities and persons with reduced mobility and, where relevant, representatives of persons with disabilities and persons with reduced mobility.
- 4. Carriers offering single multimodal contracts and multimodal hub managers shall publish the access rules referred to in paragraph 1 and provide them, upon request, in accessible format.
- 5. Where a carrier requires that a disabled person or person with reduced mobility be accompanied by another person who is capable of providing the assistance required in accordance with paragraph 2, the accompanying person shall be entitled to travel free of charge and to be seated, where practicable, next to the person with disabilities or to the person with reduced mobility.
- 6. When a carrier or intermediary makes use of the derogation provided for in Article 12 (2), it shall, upon request, inform in writing or, where necessary, in any accessible format the person with disabilities or person with reduced mobility concerned of the reasons for making use of that derogation within five working days of the refusal to accept the reservation or to issue the ticket or of the imposition of the condition of being accompanied. The carrier or intermediary shall make reasonable efforts to propose acceptable alternative transport to the person in question taking into account his or her accessibility needs.

# Information on accessibility to persons with disabilities and persons with reduced mobility

Carriers and intermediaries offering transport contracts on behalf of one or more carriers, and multimodal hub managers, shall provide persons with disabilities and persons with reduced mobility with information on the accessibility of the multimodal hub and associated facilities and of services. This information shall be provided upon request in accessible format.

#### Article 14

#### Assistance to persons with disabilities and persons with reduced mobility

In the context of single multimodal contracts, carriers, terminal managers and intermediaries shall cooperate to provide assistance free of charge to persons with disabilities and persons with reduced mobility, in accordance with the access rules referred to in Article 12(1), and offer a single notification mechanism, in accordance with the following:

(a) assistance shall be provided where the carrier, the intermediary with which the single multimodal contract was purchased, the terminal manager or the Single Point of Contact referred to in Article 15, where applicable, is notified of the passenger's need for such assistance at least 48 hours before the assistance is

- needed; a single notification per journey shall be required; the notification shall be forwarded to all carriers, terminal managers and Single Points of Contact involved in the journey;
- (b) the notification shall be accepted without additional costs, irrespective of the means of communication being used;
- (c) carriers, terminal managers and intermediaries shall take all measures necessary for the reception of notifications; where ticket vendors are unable to process such notifications, they shall indicate alternative points of purchase or alternative means to make the notification;
- (d) if the notification is made in accordance with point (a), carriers and terminal managers shall provide assistance in such a way that the person is able to take the transport services for which he or she holds a reservation as part of the single multimodal contract;
- (e) if the notification is not made in accordance with point (a), or no such notification has been made, the carriers and terminal managers shall make all reasonable efforts to provide assistance in such a way that the person with disabilities or person with reduced mobility may travel;
- (f) an assistance dog shall be permitted to accompany a person with disability or person with reduced mobility in accordance with any relevant Union<sup>16</sup> or national law.

## Single points of contact for assistance at multimodal passenger hubs

- 1. Member States shall ensure that terminal managers and carriers on their territory cooperate to establish and to operate single points of contact for persons with disabilities and persons with reduced mobility at multimodal passenger hubs in the urban nodes referred to in Annex I. The terms for the operation of the single points of contact shall be established in the access rules referred to in Article 12(1). Those single points of contact shall have the responsibility to:
  - (a) accept requests for assistance at terminals;
  - (b) communicate individual requests for assistance to terminal operators and carriers.
- 2. Member States may require that terminal managers and carriers on their territory cooperate to establish and to operate single points of contact at other multimodal passenger hubs than those referred to in paragraph 1.

#### Article 16

## Compensation in respect of mobility equipment, assistive devices and assistance dogs

1. Where terminal managers and carriers assisting persons with disabilities and persons with reduced mobility from one transport service to a connecting transport service,

Regulation (EU) No 576/2013 of the European Parliament and of the Council of 12 June 2013 on the non-commercial movement of pet animals and repealing Regulation (EC) No 998/2003 Text with EEA relevance (OJ L 178, 28.6.2013, p. 1.

either in the context of a single multimodal contract or at a multimodal passenger hub, cause the loss of, or damage to, mobility equipment, including wheelchairs, and assistive devices, or the loss of, or injury to, assistance dogs used by persons with disabilities and persons with reduced mobility, they shall be liable for that loss, damage or injury, and provide compensation without undue delay. That compensation shall comprise:

- (a) the cost of replacement or repair of the mobility equipment or assistive devices lost or damaged;
- (b) the cost of replacement or the treatment of the injury of an assistance dog that was lost or injured;
- (c) reasonable costs of temporary replacement for mobility equipment, assistive devices or assistance dogs where such replacement is not provided or is not to be provided, by the carrier or terminal manager in accordance with paragraph 2.
- 2. Where paragraph 1 applies, carriers and terminal managers shall rapidly make all reasonable efforts to provide immediately needed temporary replacements for mobility equipment or assistive devices. The person with disabilities or the person with reduced mobility shall be permitted to keep that temporary replacement equipment or device until the compensation referred to in paragraph 1 has been paid.
- 3. Where a carrier or terminal manager pays compensation under paragraph 1, no provision of this Regulation may be interpreted as restricting their right to seek compensation from any person, including third parties, in accordance with the applicable law.

#### **CHAPTER V**

## SERVICE QUALITY AND COMPLAINTS

## Article 17

#### Service quality standards

- 1. Carriers offering single multimodal contracts shall establish service quality standards and implement a quality management system to maintain service quality. The service quality standards shall at least cover the items listed in Annex II.
- 2. Carriers offering single multimodal contracts shall monitor their performance as reflected in the service quality standards. They shall publish a report on their service quality performance on their website by [2 years after the day of application of this Regulation], and every two years thereafter. That report shall not contain personal data.
- 3. Multimodal hub managers located in a Member State shall establish service quality standards based on the relevant items listed in Annex II. They shall monitor their performance pursuant to those standards and provide access to the information on their performance to the national public authorities on request.

## **Complaints**

- 1. Each carrier offering single multimodal contracts or combined multimodal tickets, each intermediary offering combined multimodal tickets and each multimodal hub manager shall set up a complaint-handling mechanism for the rights and obligations covered by this Regulation in their respective fields of responsibility. They shall make their contact details and working language, or languages, widely known to passengers.
- 2. Details of the complaint-handling procedure shall be accessible to the public, including to persons with disabilities and to persons with reduced mobility. That information shall be available upon request in the official language or languages of the Member State in which the carrier, intermediary or multimodal hub manager is operating.

#### **CHAPTER VI**

#### INFORMATION AND ENFORCEMENT

#### Article 19

## Information to passengers on their rights

- 1. When selling tickets for a multimodal journey, carriers and intermediaries shall inform passengers of their rights and obligations under this Regulation. In order to comply with that information requirement, they may use a summary of the provisions of this Regulation prepared by the Commission in all official languages of the Union and made available to the public.
- 2. Carriers and intermediaries offering multimodal journeys shall also inform passengers of the contact details of the body or bodies designated by Member States pursuant to Article 20(1), on board and on their website.
- 3. Carriers and intermediaries shall provide the information under this Article, in either paper or electronic format, or by any other means, including in accessible formats. In the event of cancellation, missed connection or long delay, they shall specify where such information can be obtained.

## Article 20

### National enforcement body

- 1. Each Member State shall designate a body or bodies responsible for the enforcement of this Regulation. To this effect, Member States may designate a body which is already responsible for the enforcement of the Union law on passenger rights. Each body shall take the measures necessary to ensure that the rights of passengers are respected.
- 2. Member States shall inform the Commission of the body or bodies designated in accordance with this Article and of its or their respective responsibilities. The Commission and the bodies designated shall publish that information on their websites.

3. By 1 June XXXX [2 years after the Regulation becomes applicable] and every 2 years thereafter, the national enforcement bodies shall publish a report on their website on their activity in the previous 2 calendar years, containing in particular a description of actions taken in order to implement this Regulation and statistics on complaints and sanctions applied.

#### Article 21

## Complaint handling by national enforcement bodies and other bodies

- 1. Member States shall designate the national enforcement body or any other body as the entity responsible for the handling of complaints under paragraph 2 of this Article.
- 2. Where different bodies are designated under paragraph 1 and Article 20(1), reporting mechanisms shall be set up to ensure the exchange of information between them in order to help the national enforcement body to carry out its tasks of supervision and enforcement, and in order to enable the complaint-handling body designated under this paragraph to collect the information necessary to examine individual complaints.
- 3. Without prejudice to the possibility for consumers to seek alternative redress pursuant to Directive 2013/11/EU, after having complained unsuccessfully to the carrier, intermediary or multimodal hub manager pursuant to Article 18, passengers may complain about an alleged infringement of this Regulation to the body designated under paragraph 1.
- 4. Passenger complaints about an incident in the context of a single multimodal contract shall be handled by the body designated under paragraph 1 of the Member State where the contracting carrier is established.
- 5. Passenger complaints about an incident in the context of a combined multimodal ticket offered by carriers shall be handled by the body designated under paragraph 1 of the Member State where the carrier offering this ticket is established or, if the carrier is established outside of the Union, of the Member State where the legal or natural person acting as its legal representative is based. If the carrier does not have such a representative, the complaints shall be handled by the body designated under paragraph 2 of the Member State where the journey starts or ends.
- 6. Without prejudice to paragraphs 3 and 4, passenger complaints about an intermediary shall be handled by the body designated under paragraph 1 of the Member State where the intermediary is established or, if the intermediary is established outside of the Union, of the Member State where the legal or natural person acting as its legal representative is based. If the intermediary does not have such a representative, the complaints shall be handled by the body designated under paragraph 2 of the Member State where the journey starts or ends.
- 7. Where a complaint relates to alleged infringements by multimodal hub managers, the complaint shall be handled by the body designated under paragraph 1 of the Member State on whose territory the incident occurred.

## Article 22

## Risk-based approach to the monitoring of compliance with passenger rights

1. The national enforcement bodies shall develop a compliance monitoring programme to monitor compliance with the obligations laid down in this Regulation on the basis

- of a risk assessment. The programme shall allow for the detection and correction of recurrent non-compliance in the implementation of passenger rights. The compliance monitoring programme shall include a representative sample of monitoring activities.
- 2. The risk assessment shall be based on a factual assessment that takes into account complaints made by passengers to these bodies, where available, findings of the monitoring activities carried out by these bodies, information referred to in Article 23 (1) and (3), as well as other sources of information pertaining to the application of this Regulation on the territory of the respective Member State.
- 3. The risk assessment shall be carried out for the first time by 30 June XXXX [1 year after the Regulation becomes applicable], and every two years thereafter.
- 4. Compliance monitoring activities shall be based on the assessment of risks and carried out by means of audits, inspections, interviews, verifications and examination of documents as appropriate. They shall include both announced and unannounced activities. The monitoring activities shall be proportionate to the risks identified.
- 5. National enforcement bodies shall ensure the swift rectification of non-compliance by carriers and terminal managers as identified during their monitoring activities. They shall require carriers to submit an action plan to remedy non-compliance, as appropriate.
- 6. The compliance monitoring programme under paragraph 1, the risk assessment under paragraph 2 as well as their findings shall be communicated to the Commission by 30 June XXXX [2 years after the Regulation becomes applicable], and every two years thereafter. Where applicable, they shall be included in the reports referred to in Article 20 (3).

## Sharing of information with national enforcement bodies

- 1. Carriers, intermediaries, terminal managers and multimodal hub managers shall provide the national enforcement bodies with relevant documents and information at their request without undue delay and, in any event, within one month from the receipt of the request.
- 2. In complex cases, the national enforcement body may extend this period to a maximum of three months from the receipt of the request.
- 3. In carrying out their functions, the national enforcement bodies shall take account of the information submitted to them by the body designated to handle complaints, if this is a different body.

#### Article 24

# **Cooperation between Member States and the Commission**

- 1. The Member States shall regularly send relevant information concerning the application of this Regulation to the Commission, which will make this information available in electronic form to other Member States.
- 2. At the request of the Commission, the national enforcement bodies shall investigate specific suspected practices of non-compliance with the obligations laid down in this Regulation by one or several carriers, terminal managers and intermediaries and report its findings to the Commission within four months of the request.'

#### **CHAPTER VII**

#### FINAL PROVISIONS

#### Article 25

#### **Penalties**

Member States shall lay down the rules on penalties applicable to infringements of this Regulation and shall take all measures necessary to ensure that they are implemented. The penalties provided for shall be effective, proportionate and dissuasive. Member States shall notify the Commission of those rules and measures and shall notify it without delay of any subsequent amendment affecting them.

#### Article 26

## **Committee procedure**

- 1. The Commission shall be assisted by the committee established under Article 38 of Regulation (EU) 2021/782.
- 2. Where reference is made to this paragraph, Article 5 of Regulation (EU) No 182/2011 shall apply.

#### Article 27

## Report

By XXX [five years after the date of application pursuant to Art. 28], the Commission shall report to the European Parliament and the Council on the implementation and the results of this Regulation.

The report shall be based on information to be provided pursuant to this Regulation.

#### Article 28

## Entry into force and application

This Regulation shall enter into force on the twentieth day following that of its publication in the *Official Journal of the European Union*.

It shall apply from [1 year after entry into force].

This Regulation shall be binding in its entirety and directly applicable in all Member States. Done at Brussels,

For the European Parliament The President For the Council The President



Brussels, 29.11.2023 COM(2023) 752 final

ANNEXES 1 to 2

## **ANNEXES**

## to the

Proposal for a Regulation of the European Parliament and of the Council on passenger rights in the context of multimodal journeys

 $\{ SEC(2023) \ 392 \ final \} - \{ SWD(2023) \ 386 \ final \} - \{ SWD(2023) \ 387 \ final \} - \{ SWD(2023) \ 389 \ final \}$ 

EN EN

## ANNEX I

#### LIST OF URBAN NODES

## Belgium

Antwerpen

Arlon

Brugge

Bruxelles/Brussel

Charleroi

Gent

Hasselt

Leuven

Liège

Namur

Ottignies-Louvain-la-Neuve

# Bulgaria

Burgas

Plovdiv

Ruse

Sofia

Stara Zagora

Varna

Vidin

## **Czech Republic**

Brno

České Budějovice

Hradec Králové

Liberec

Olomouc

Ostrava

Pardubice

Plzeň

Praha

Ústí nad Labem

### **Denmark**

Aalborg

Århus

København

Odense

## Germany

Aachen

Augsburg

Berlin

Bielefeld

Bochum

Bonn

Bottrop

Braunschweig

Bremen

Bremerhaven

Chemnitz

Darmstadt

Dortmund

Dresden

Duisburg

Düsseldorf

Erfurt

Erlangen

Essen

Frankfurt am Main

Freiburg im Breisgau

Fürth

Gelsenkirchen

Gießen

Göttingen

Gütersloh

Hagen

Halle (Saale)

Hamburg

Hamm

Hannover

Heidelberg

Heilbronn

Herne

Hildesheim

Ingolstadt

Jena

Kaiserslautern

Karlsruhe

Kassel

Kiel

Koblenz

Köln

Krefeld

Landshut

Leipzig

Leverkusen

Lübeck

Ludwigshafen am Rhein

Magdeburg

Mainz

Mannheim

Mönchengladbach

Mülheim an der Ruhr

München

Münster

Stuttgart Trier Ulm Wiesbaden Wolfsburg Wuppertal Würzburg Estonia Tallinn Tartu **Ireland** Corcaigh/Cork Baile Átha Cliath/Dublin Gaillimh/Galway Greece EN EN 4

Neuss

Nürnberg

Oberhausen

Osnabrück Paderborn

Pforzheim Potsdam

Recklinghausen

Regensburg

Saarbrücken

Rostock

Siegen

Solingen

Offenbach am Main

Oldenburg (Oldenburg)

Alexandroupolis Athína Chalkida Chania Heraklion Ioannina Kerkyra Kozani Lamia Larisa Mytilini **Patras** Rodos Syros Thessaloniki Volos **Spain** A Coruña Albacete Algeciras Alicante Almería Badajoz Barcelona Metropolitan Area Bilbao Metropolitan Area Burgos Caceres Cádiz Cartagena Castellón Ceuta Córdoba

Elche Ferrol Gijón Gerona Granada Huelva Jaén Jerez Las Palmas de Gran Canaria Metropolitan Area León Lleida/Lérida Logroño Lugo Madrid Metropolitan Area Málaga Marbella Melilla Murcia Ourense Oviedo Palma de Mallorca Pamplona Salamanca San Sebastián-Donostia Santa Cruz de Tenerife Metropolitan Area Santander Santiago de compostela Sevilla Metropolitan Area Tarragona-Reus Metropolitan Are Valencia

Vitoria-Gasteiz Zaragoza

Valladolid

Vigo

### **France**

Aix-en-Provence

Ajaccio (Corse)

Amiens

Angers

Annecy

Besançon

Bordeaux

**Brest** 

Caen

Cayenne

Clermont-Ferrand

Dijon

Grenoble

Guadeloupe

Le Havre

Le Mans

Lille

Limoges

Lyon Metropolitan Area

Martinique

Marseille

Mayotte

Metz

Montpellier

Mulhouse

Nancy

Nantes

Nice

Nimes

Orléans

Paris Metropolitan Area

Perpignan

**Poitiers** 

Réunion

Reims

Rennes

Rouen

Saint-Étienne

Strasbourg

Toulon

Toulouse

Tours

## Croatia

Osijek

Rijeka

Split

Varaždin

Zagreb

## Italy

Ancona

Andria

Aosta

Bari

Bergamo

Bologna

Bolzano

Brescia

Cagliari

Campobasso

Catania

Ferrara

Firenze

Foggia

Forlì

Genova

Latina

Lecce

Livorno

Messina

Milano

Modena

Monza

Napoli

Novara

Padova

Palermo

Parma

Perugia

Pescara

Piacenza

Potenza

Prato

Ravenna

Reggio Calabria

Reggio Emilia

Rimini

Roma

Salerno

Sassari

Siracusa

Taranto

Terni

Torino

Trento

Trieste

Udine

Venezia

Verona

Vicenza

# **Cyprus**

Lefkosia

Lemesos

Larnaka

## Latvia

Rīga

## Lithuania

Kaunas

Klaipėda

Panevėžys

Šiauliai

Vilnius

# Luxembourg

Luxembourg

# Hungary

Budapest

Debrecen

Győr

Kecskemét

Miskolc

Nyíregyháza

Pécs

Szeged

Székesfehérvár

## Malta

Valletta

### **Netherlands**

Alkmaar

Almere

Amersfoort

Amsterdam

Apeldoorn

Arnhem

Breda

Den Bosch

Den Haag

Dordrecht

Eindhoven

Emmen

Enschede

Groningen

Haarlem

Leeuwarden

Leiden

Maastricht

Middelburg

Nijmegen

Rotterdam

Tilburg

Utrecht

Venlo

Zoetermeer

Zwolle

### Austria

Bregenz

Eisenstadt

Graz

Innsbruck

Klagenfurt

Linz-Wels

Salzburg

Sankt Pölten

Wien

### **Poland**

Białystok

Bielsko-Biała

Bydgoszcz

Częstochowa

Elbląg

Gorzów Wielkopolski

Kalisz

Katowice / Górnośląska Metropolis

Kielce

Koszalin

Kraków

Legnica

Lublin

Łódź

Olsztyn

Opole

Płock

Poznań

Radom

Rybnik

Rzeszów

Szczecin- Świnoujście

Tarnów

Toruń

Tricity

Wałbrzych

Warszawa

Włocławek

Wrocław

Zielona Góra

## **Portugal**

Barcelos

Braga

Coimbra

Évora

Faro-Loulé

Guimarães

Madeira

Leiria

Lisboa Metropolitan Area

Ponta Delgada (Açores)

Porto Metropolitan Area

Vila Nova de Famalição

Viseu

### Romania

Arad

Bacău

Baia Mare

Botoșani

Brăila

Brașov

București

Buzău

Cluj-Napoca

Constanța Craiova Galați Iași Oradea

Piatra Neamt

Pitești

Ploiești

Râmnicu Vâlcea

Satu Mare

Sibiu

Târgu Mureș

Timișoara

## Slovenia

Ljubljana

Maribor

### Slovakia

Bratislava

Košice

Nitra

Žilina

## **Finland**

Helsinki

Jyväskylä

Kuopio

Lahti

Oulu

Tampere

Turku-Naantali

## Sweden

Borås

Eskilstuna

Gävle

Göteborg

Halmstad

Helsingborg

Jönköping

Linköping

Lund

Malmö

Norrköping

Örebro

Södertälje

Stockholm

Sundsvall

Umeå

Uppsala

Västerås

#### **ANNEX II**

#### MINIMUM SERVICE QUALITY STANDARDS

Information to passengers on their rights as required by this Regulation (in a transparent, accessible, non-discriminatory, effective and proportionate manner) and tickets

Punctuality of services, and general principles to cope with disruptions to services

Number of missed connections

Percentage of missed connections

Cleanliness of the means of passenger transport and the terminal facilities (air quality and temperature control inside, hygiene of sanitary facilities, etc.)

Results of the customer satisfaction survey(s)

#### Complaint handling

- i. the number of complaints,
- ii. types of complaints and
- iii. the outcome of complaints,
- iv. reimbursements and
- v. compensation for non-compliance with service quality standards

Assistance provided to persons with disabilities and persons with reduced mobility (PRM)

- i. number of denied transport services for a PRM, and the reason for it;
- ii. their recognised assistance dog or their mobility equipment and the reasons for it,
- iii. number of cases related to the loss or damages of mobility equipment and their outcome,
- iv. number of cases where the carrier obliged a person with disability or reduced mobility to be accompanied by someone to comply with the safety requirements,
- v. discussions concerning this assistance with representative organisations and, where relevant, representatives of PRM
- vi. the implementation of the disability awareness and assistance training